Author's response to reviews

Title: A multidisciplinary clinical treatment of locally advanced rectal cancer complicated by rectovesical fistula

Authors:

Tiancheng Zhan (jessecrash@sina.com.cn)
Lin Wang (wanglinmd@gmail.com)
Ming Li (limingmd@126.com)
Zhongwu Li (zhwuli2008@sina.com)
Yong Cai (caiyong109@souhu.com)
Lin Shen (lin100@medmail.com.cn)
Jin Gu (zlguj@yahoo.com.cn)

Version: 2 Date: 10 May 2012

Author's response to reviews: see over
Dear Patel:
I am submitting my paper entitled “A multidisciplinary clinical treatment of locally advanced rectal cancer complicated by rectovesical fistula” for your consideration as an original case report for BMC Cancer (Journal of medical, Case reports). The report is to call attention to the importance of MDT in complicated rectal cancer cases.

The case is a rectal cancer patient with rectovesical fistula who benefited from multimodality treatment (MDT). Rectal cancer with rectovesical fistula is a complicated rare situation being difficult to deal with highlighting MDT’s function. For the case our MDT panel discussed and decide the treatment as neoadjuvant radiochemotherapy, total pelvic excision and adjuvant chemotherapy. The postoperative pathology result is pathological complete response (PCR) which suggest well prognosis and we now anticipate the long term survival.

This paper has not been submitted elsewhere for publication or is under consideration of publication. We appreciate your attention to our manuscript. And we are looking forward to your reply at earliest convenience.

Sincerely,

Zhan Tiancheng
Key Laboratory of Carcinogenesis and Translational Research (Ministry of Education), Department of Colorectal Surgery, Peking University School of Oncology, Beijing Cancer Hospital & Institute, Beijing, China
Email: jessecrash@sina.com; Tel: 13911377261; Fax: 86-10-8814-1032.

Jin Gu, M.D., FACS. Professor of Colorectal Surgery, Peking University School of Oncology, Beijing Cancer Hospital
52 Fu Cheng Lu, Haidian District, Beijing 100142 PR China
e-mail: zlguj@yahoo.com.cn, Tel: 86-10-8814-1032; Fax: 86-10-8814-1032